Review article

Insights into methotrexate in rheumatoid arthritis: a clinical review

Genga EK^{1,2}, Oyoo GO^{1, 2}, Ezzi MS¹

Abstract

¹Department of Clinical Medicine and Therapeutics, College of Health Sciences, University of Nairobi, P. O. Box 19676-00202, Nairobi, Kenya ²Nairobi Arthritis Clinic

Corresponding author:

Dr EK Genga, Department of Clinical Medicine and Therapeutics, College of Health Sciences, University of Nairobi, P. O. Box 19676-00202, Nairobi, Kenya. Email: eugenekalman@ gmail.com **Objective**: To review the efficacy and safety profile of methotrexate in rheumatoid arthritis.

Data source: Published original research work and reviews were searched in English related to efficacy and safety profile of methotrexate in rheumatoid arthritis.

Study design: Only articles that emphasis on efficacy and safety profile of methotrexate in rheumatoid arthritis.

Data extraction: Online and library searches done.

Data synthesis: Data added and summarized.

Conclusion: Methotrexate (MTX) has been the mainstay of treatment of patients with Rheumatoid Arthritis (RA). It has been used for over 50 years as the backbone in the treatment in a number of rheumatic diseases and thus it remains a gold standard of therapy for RA. Clinical trial results indicate that weekly low dose MTX is generally safe and effective in the treatment of RA. Factors that favour a good response to MTX are male gender, non-smoking, early disease stage, absence of previous DMARD use, lower baseline disease activity, concomitant corticosteroids, inflammatory biomarkers (TNF- α levels, ESR, CRP) and HLA-DRB1 shared epitope (SE)-negativity. Folate supplementation has been shown to reduce the risk of adverse events.

Key words: Methotrexate, Rheumatoid arthritis, Efficacy, Safety profile

Introduction

Methotrexate was first used for the treatment of rheumatoid arthritis in 1951¹. It emerged at the same time as glucocorticoids. It received little attention in the therapy of rheumatic diseases until retrospective reports appeared about 30 years later^{2,3}. The initial short term studies in the mid-1980s demonstrated that it was more potent, efficacious and superior to placebo in patients with chronic severe RA. Long-term open prospective studies have since showed that the response is sustained and that toxicity is manageable. These insights have led to the evolution of methotrexate into its present status as the main stay of treatment of patients with RA. However, there is a large variability in the medical practice regarding the use of methotrexate for RA. The variability occurs mainly with regards to the starting dose, adjusting dosages, routes of administration and concomitant use of folic or folinic acid. The purpose of this article is to provide guidance on the use of methotrexate for the treatment of RA.

Mechanism of action

Methotrexate was introduced for the therapy of rheumatoid arthritis without any clear understanding of its mechanism of action. With methotrexate, studies are even more difficult to interpret because the effects of methotrexate are observed over weeks to months in patients. Another caveat is that studies in animals may be misleading because the doses of the drug used are not similar to those used in patients. It's important to note that the methotrexate effect in animals is seen over a shorter duration of time as compared to humans⁴. Despite these, there are a number of postulated mechanisms of action.

The first hypothesis, based on methotrexate's known anti-folate properties. Methotrexate is a structural analogue of folic acid that competitively inhibits the binding of dihydrofolic acid to the enzyme dihydrofolate reductase. Dihydrofolate reductase is enzyme responsible for reducing dihydrofolic acid to the active metabolite, folinic acid. Therefore, methotrexate decreases the amount of intracellular folinic acid available and affects the intra-cellular acid dependent folinic metabolic pathways. These pathways include purine and pyrimidine metabolism, as well as amino acid and polyamine synthesis. Its postulated by inhibiting these methylated reactions it inhibits proliferation of the pro-inflammatory cells and cytokines responsible for synovial inflammation in RA^{5,6}.

Other mechanism of action includes increasing T cell apoptosis and release of endogenous anti-inflammatory adenosine. It also alters the expression of cellular adhesion molecules thus reducing in expression of cellular adhesion molecules and anti-angiogenesis effects via indirect mechanisms such as disruption of macrophage interaction^{7,8}.

Efficacy for rheumatoid arthritis

Methotrexate can be given by oral, intramuscular, and subcutaneous routes. The oral route has a variable absorption and has less serum levels when compared to parenteral administration ^{9, 10}. This may have therapeutic implications, as some patients may seem to respond better to parenteral therapy, presumably because more drugs reaches the circulation especially when higher doses within the therapeutic range are used. The liquid MTX formulation (prepared for parenteral use) may be consumed orally by mixing it with juice; this preparation is cheaper than tablets and may be used if expense is a serious issue and if the patient can be relied upon to measure the precise volume of drug. It is not recommended for patients with decreased finger dexterity, limited vision, or impaired cognition. Research has shown that better disease activity scores can be attained from switching to parenteral administration with patients on maximally tolerated oral doses of MTX. Toxicity has been noted to be higher in the parental group^{11,12}.

Its use in rheumatoid arthritis is well known. Apart from improving disease activity scores, it also has benefits of reducing mortality, improving quality of life and reducing radiographic joint damage^{13,14}. With the introduction of biologics, data from Kleinert's study, Gruppo Italiano Studio Early Arthritis (GISEA), British Society for Rheumatology Biologics Register (BSRBR), and Research in Active Rheumatoid Arthritis (ReAct) still show that methotrexate is still the anchor for rheumatoid arthritis management¹⁵⁻¹⁷. Despite its popularity in rheumatoid arthritis, the dose and route of administration has been varied. To determine the best dose and route of administration, we looked into data from previous studies.

The starting dose has been an area of contention. Furst *et al*¹⁸ compared starting doses of 5-10mg weekly, 12.5-20mg weekly and 25-35mg weekly vs placebo in RA patients who had the disease for a long duration and had failed other Disease Modifying Antirheumatic Drugs (DMARDs). The best response in terms of tender joint counts, pain and global status versus the placebo was noted in the 12.5-20mg group. The 25-35mg group had the highest adverse events as measured by gastrointestinal and mucocutaneous toxicities.

In early RA DMARD naïve patients, Verstappen *et* al^{19} looked at two groups. First group started at 7.5mg per week and increased it per month by 5mg up to a maximum of 25mg weekly. The second group had a slower increment of 5mg every 3 months to a mean of 18mg weekly. Patients in the group with the fast increment achieved better disease control as measured using tender and swollen joint counts, pain and global status. Interestingly both groups had similar records of adverse events. The general recommendations are start patients on 15 mg weekly orally, escalating 5 mg per month to 25-30 mg weekly as tolerated depending on disease activity, size, and age of the patient, the presence of comorbidities, and renal function. If response is inadequate, one should consider switching to SC where available²⁰.

The response to any DMARDs be it synthetic or biologic varies from one RA patient to another. This is

due to the unique and complex pathogenesis of RA which causes varied clinical presentation. The next frontier in management is being able to predict the responders to drugs used in RA treatment. A novel biomarker of response has been methotrexate polyglutamate. These are the active MTX metabolites that produce the anti-inflammatory effects and their levels have been found to correlate to disease activity^{21,22}. Higher levels are associated with good responses, while lower levels might indicate the need for either more MTX, or the possibilities of poor adherence to the treatment regimen or poor absorption of the dose inhibit enzymes of folate metabolism^{21,22}. Data is still limited and more research is needed before this can be applied to routine clinical use. The known clinical and biological factors that predict good response to MTX are male gender, non-smoking, early disease stage, absence of previous DMARD use, lower baseline disease activity, concomitant corticosteroids, inflammatory biomarkers (TNF- α levels, ESR, CRP) and HLA-DRB1 shared epitope (SE)-negativity²²⁻²³.

Safety profile

MTX use has been associated with a variety of adverse effects. The range of severity is influenced by the MTX dose and treatment regimen. The major side effects are reviewed here.

Common minor adverse events

Gastrointestinal

The common side effects include nausea, loose stools and stomatitis. Stomatitis can occur at any dose of MTX but is more likely to be seen at higher doses. These are mainly due to sub-optimal supplementation of folic acid. Patients with mild to moderate stomatitis, one can gradually increase the folic acid dose from 1 mg daily increments up to a maximum of 5 mg until the toxicity is controlled. Patients presenting with severe oral ulcers may require both a lowering of the MTX dose and an increase of folic acid²⁴.

Neurotoxicity

Manifestations include severe headaches, fatigue, and problems in concentrating which may require reducing the MTX dose or discontinuation in some patients. The exact mechanism is still unclear. Some have suggested it may be related to the accumulation of adenosine due to the inhibition of purine synthesis²⁵. **Fever**

MTX can induce fever either directly or indirectly through infections. Infections are rare but commonly occur with co-administration of glucocorticoids, azathioprine and tumour necrosis factor inhibitors. Pneumocystis jiroveci, herpes zoster and fungi can occur. Infections are a common cause of drug withdrawal among those administered MTX for prolonged periods²⁶.

Potential severe adverse effects

Hepatotoxicity

Low dose methotrexate can have adverse effects on the liver. These include reversible transaminase elevation (most common), liver fibrosis and liver cirrhosis (rare). Risk factors include alcohol and preexisting liver disease. Other risk factors for elevated transaminases include obesity, untreated high cholesterol, Aspartate Transaminase (AST) or Alanine Transaminase (ALT) elevations above the upper limit of normal at baseline (before starting MTX) use of a biologic agent in addition to the MTX, and lack of folic acid supplementation²⁷. Some authors suggest that folate supplementation may help prevent MTX induced hepatotoxicity. Methotrexate depletes the folate hepatic stores and supplementation either folic acid 1 mg per day or folinic acid 2.5 mg per week is associated with a reduced incidence of serum transaminase elevation²⁸. However, more research is needed to establish the relationship between folate depletion and hepatic toxicity.

Data from the CORRONA database that 1953 RA and 151 psoriatic arthritis patients, showed there was an increased risk of transaminase elevations with a combination of MTX and leflunomide compared with either drug used alone²⁹. The overall incidence of elevations in aminotransferase enzymes in patients with RA receiving MTX, leflunomide, MTX plus leflunomide, and neither was 22, 17, 31, and 14%, respectively. Elevations > 2 x ULN occurred in 1-2% of patients on MTX or LEF monotherapy compared to 5% with the combination. After multivariable adjustment and compared with either monotherapy, combination MTX + LEF was associated with greater risk according to MTX dose used as part of the combination.

Data from the SMILE cohort (Safety of Methotrexate and Leflunomide in RA trial) over a period of 12 months examined transaminase abnormalities in 2975 patients. The overall incidence of elevations in aminotransferase enzymes in patients receiving MTX (52.2%), Leflunomide (7.3%), MTX plus Leflunomide (13.9%), and neither (26.6%) was 12, 16, 19, and 14%, respectively³⁰. No reports of liver fibrosis or cirrhosis were recorded. Only a small number of patients had to stop taking MTX and/or LEF cessation due to AEs.

The above results are reassuring that rarely do serious liver abnormalities occur in patients using low dose. Other novel non-invasive methods are under investigation to detect hepatic injury and fibrosis. As an example, ultrasound-based transient elastography has been used in preliminary studies of patients receiving MTX for inflammatory arthritis, psoriasis, and gastroenterologic disorders but has not been evaluated for its utility in monitoring patients with RA receiving MTX in clinical practice³¹.

Recommendations dictate that, in patients on a stable dose of MTX, monitoring at an interval of every 8 to 12 weeks is appropriate after three months of therapy and monitoring every 12 weeks can be performed beyond six months of therapy. Liver biopsies are no longer recommended on all patients on MTX but only done based upon the presence of risk factors for hepatotoxicity³². A pre-treatment biopsy is considered only for patients

with a history of excessive prior alcohol consumption, persistently abnormal baseline AST values, or chronic hepatitis B or C infection ³².

Pulmonary abnormalities

The incidence of pulmonary abnormalities remains low. A systematic review on 21 prospective studies reported that only 15 (0.43%) out of 3463 RA patients on MTX treatment on follow up for 36.5 months developed MTX pneumonitis³³. The authors concluded MTX pneumonitis be considered as an acute hypersensitivity reaction, occurs early in the course of MTX, thus it does not seem to be a problem of long-term treatment by MTX. Factors that have been associated with MTX- induced lung injury include higher weekly doses of methotrexate, preexisting interstitial lung diseases, abnormal pulmonary function tests prior to therapy, decreased elimination of methotrexate (e.g., as seen in renal insufficiency or with the presence of third-space fluid collections such as ascites), hypoalbuminemia (either before or during therapy), diabetes mellitus previous use of diseasemodifying antirheumatic drugs, cigarette smoking, and low body weight³⁴. The mechanism by which some of these factors may confer excess risk is unclear. There is data supporting hyperinsulinemia, which may occur with treatment for diabetes mellitus, is associated with increased polyglutamation of methotrexate. The previous use of disease-modifying antirheumatic drugs may be a marker for more severe rheumatoid arthritis, and hypoalbuminemia could potentially result in a lower degree of protein binding and higher free levels of methotrexate³⁴⁻³⁶. The acute presentation typically includes fever, chills, malaise, nonproductive cough, dyspnea, and chest pain: the subacute presentation is characterized by a more insidious onset of dyspnea, cough, and fever. The majority of patients who develop methotrexate pulmonary toxicity do so within the first year of therapy³³.

Myelosuppression

This is commonly seen when using MTX in high doses. With low dose MTX therapy, anaemia, neutropenia and lymphopenia are the commonest abnormalities encountered in RA. Thrombocytopenia is rare now that Felty's syndrome isn't common³⁷. A more serious problem is pancytopenia which has been associated with elderly, those with concomitant use of dihydrofolate reductase inhibitors, and patients with renal impairment³⁸.

In a study by Bird *et al*³⁰ where they reported a lower incidence of neutropenia rate of neutropenia was higher in patients not taking MTX, than those taking MTX as monotherapy. They defined neutropenia as neutrophil count of $< 2.0 \times 10^{9}$ /L and found that 2.3% of the MTX monotherapy group, 5.5% of the LEF monotherapy group, 3.9% of the MTX/LEF combination group and 4.2% of the group taking neither drug. They however concluded that these values did not correspond to an increased incidence of infection.

Guidelines recommend that a routine peripheral complete blood count should be performed every four weeks during the first three months of therapy, every 8 to 12 weeks from three to six months, and every 8 to 12 weeks thereafter, depending upon the nature and/or severity of abnormalities noted during monitoring³².

Risk of malignancy

The incidence of cancer and mortality by cancer are slightly higher in RA cohorts as compared to the general population^{39,40}. Hematopoietic and lung cancers make up the majority of numbers³⁹. Rarely, lymphoproliferative "malignancies" may develop after long-term therapy but regress spontaneously after MTX is withdrawn. They are usually of B-cell origin and some are associated with latent Epstein Barr virus infection⁴¹. Chemotherapy should be withheld until MTX has been stopped, since some of these tumours regress within four weeks after MTX has been discontinued. Continued vigilance is necessary in those who regress, since relapse has been reported^{42,43}. Drug therapy does not confer a direct risk to developing cancer. A Canadian study of 23810 patients followed up between 1980 - 2003 reported 619 haematological malignancies (lymphoma in 346, leukaemia in 178 and myeloma in 95). Analysis performed to assess the effect of DMARD therapy showed that the unadjusted ratios for haematologic malignancy after drug exposures were: MTX 1.18 (95% CI 0.99-1.40), azathioprine 1.44 (95% CI 1.01-2.03) and cyclophosphamide 2.21 (95% CI, 1.52-3.20)⁴⁴. There is an increased risk for malignancy, but the above data reassures that the numbers are still low.

Conclusion

Methotrexate is still the anchor for rheumatoid arthritis management. Clinical trial results indicate that weekly low dose MTX is generally safe and effective in the treatment of RA. Factors predictors of good response to MTX are male gender, non-smoking, early disease stage, absence of previous DMARD use, lower baseline disease activity, concomitant corticosteroids, inflammatory biomarkers (TNF- α levels, ESR, CRP) and HLA-DRB1 shared epitope (SE)-negativity. It has advantages in long term treatment due to cost and is generally well tolerated due to its favourable adverse effect profile. The most commonly observed side effects of MTX at doses typically used for the treatment of RA are rarely lifethreatening. Folate supplementation has been shown to lower the risk of adverse events. Physicians need to know the risks associated with its use and monitor accordingly.

References

- 1. Gubner R, August S, Ginsberg V. Therapeutic suppression of tissue reactivity. II. Effect of aminopterin in rheumatoid arthritis and psoriasis. *Am J Med Sci.* 1951; **221**:176.
- 2. Willkens RF, Watson MA. Methotrexate: a perspective of its use in the treatment of rheumatic diseases. *J Lab Clin Med.* 1982; **100**:314.
- 3. Hoffmeister RT. Methotrexate therapy in rheumatoid arthritis: 15 years' experience. *Am J Med.* 1983; **75**:69.
- 4. Schiff MH, Jaffe JS, Freundlich B. Head-tohead, randomised, crossover study of oral versus subcutaneous methotrexate in patients with rheumatoid arthritis: drug-exposure limitations of

oral methotrexate at doses ≥ 15 mg may be overcome with subcutaneous administration. *Ann Rheum Dis.* 2014; **73**:1549.

- 5. Gerards AH, de Lathouder S, de Groot ER, *et al.* Inhibition of cytokine production by methotrexate. Studies in healthy volunteers and patients with rheumatoid arthritis. *Rheumatology* (Oxford). 2003; **42**:1189.
- 6. Genestier L, Paillot R, Fournel S, *et al.* Immunosuppressive properties of methotrexate: apoptosis and clonal deletion of activated peripheral T cells. *J Clin Invest.* 1998; **102**:322.
- Bunescu A, Seideman P, Lenkei R, Levin K, Egberg N. Enhanced Fc gamma receptor I, alphaMbeta2 integrin receptor expression by monocytes and neutrophils in rheumatoid arthritis: interaction with platelets. *J Rheumatol.* 2004; **31**: 2347-2355.
- 8. Miranda-Carús ME, Balsa A, Benito-Miguel M, Pérez de Ayala C, Martín-Mola E. IL-15 and the initiation of cell contact-dependent synovial fibroblast-T lymphocyte cross-talk in rheumatoid arthritis: effect of methotrexate. *J Immunol.* 2004; **173**: 1463-1476.
- 9. Cronstein BN. Low-dose methotrexate: a mainstay in the treatment of rheumatoid arthritis. *Pharmacol Rev.* 2005; **57**(2):163-172.
- Hamilton RA, Kremer JM. Why intramuscular methotrexate may be more efficacious than oral dosing in patients with rheumatoid arthritis. *Br J Rheumatol.* 1997; 36: 86.
- 11. Braun J, Kästner P, Flaxenberg P, Währisch J, Hanke P, *et al.* Comparison of the clinical efficacy and safety of subcutaneous versus oral administration of methotrexate in patients with active rheumatoid arthritis: results of a six-month, multicenter, randomized, double-blind, controlled, phase IV trial. *Arthritis Rheum.* 2008; **58**: 73-81.
- 12. Bakker MF, Jacobs JW, Welsing PM, *et al.* Are switches from oral to subcutaneous methotrexate or addition of ciclosporin to methotrexate useful steps in a tight control treatment strategy for rheumatoid arthritis? A post hoc analysis of the CAMERA study. *Ann Rheum Dis.* 2010; **69**:1849.
- 13. Hanrahan PS, Scrivens GA, Russell AS. Prospective long term follow-up of methotrexate therapy in rheumatoid arthritis: toxicity, efficacy and radiological progression. *Br J Rheumatol.* 1989; **28**: 147-153.
- Weinblatt ME, Maier AL, Fraser PA, Coblyn JS. Long term prospective study of methotrexate in rheumatoid arthritis: conclusion after 132 months of therapy. *J Rheumatol.* 1998; 25: 238-242.
- Kleinert S, Tony HP, Krause A, *et al.* Impact of patient and disease characteristics on therapeutic success during adalimumab treatment of patients with rheumatoid arthritis: data from a German non interventional observational study. *Rheumatol Intern.* 2011; **32**(9): 2759–2767.
- Burmester GR, Ferraccioli G, Flipo R-W, *et al.* Clinical remission and/or minimal disease activity in patients receiving adalimumab treatment in a multinational, open-label, twelve-week study. *Arthritis Care Res.* 2008; **59**(1): 32–41.

- 17. Kristensen LE, Kapetanovic MC, G⁻ulfe A, S⁻oderlin M, *et al.* Predictors of response to anti-TNF therapy according to ACR and EULAR criteria in patients with established RA: results from the South Swedish arthritis treatment group register. *Rheumatology*. 2008; **47**(4): 495–499.
- Furst DE, Koehnke R, Burmeister LF, Kohler J, Cargill I. Increasing methotrexate effect with increasing dose in the treatment of resistant rheumatoid arthritis. *J Rheumatol.* 1989; 16: 313-320.
- 19. Verstappen SM, Jacobs JW, van der Veen MJ, Heurkens AH, Schenk Y, *et al.* Intensive treatment with methotrexate in early rheumatoid arthritis: aiming for remission. Computer Assisted Management in Early Rheumatoid Arthritis (CAMERA, an open-label strategy trial). *Ann Rheum Dis.* 2007; **66**: 1443-1449.
- 20. Visser K, Katchamart W, Loza E, Martinez-Lopez JA, Salliot C, *et al.* Multinational evidence based recommendations for the use of methotrexate in rheumatic disorders with a focus on rheumatoid arthritis: integrating systematic literature and expert opinion of a broad international panel of rheumatologists in the 3E initiative. *Ann Rheum Dis.* 2008; **68**: 1086-1093.
- 21. Dervieux T, Greenstein N, Kremer J. Pharmacogenomic and metabolic biomarkers in the folate pathway and their association with methotrexate effects during dosage escalation in rheumatoid arthritis. *Arthritis Rheum.* 2006; **54**: 3095.
- 22. Halilova KI, Brown EE, Morgan SL, Bridges Jr SL, Hwang MH, Arnett DK, *et al.* Markers of treatment response to methotrexate in rheumatoid arthritis: where do we stand?" *Intern J Rheumatol.* 2012; **2012**: 1-7.
- 23. Genga EK, Moots TJ, Oyoo GO. DMARD use in rheumatoid arthritis: can we predict treatment response? *Afr J Rheumatol*. 2015; **3**(2):50-58.
- 24. Van Ede AE, Laan RF, Rood MJ, *et al.* Effect of folic or folinic acid supplementation on the toxicity and efficacy of methotrexate in rheumatoid arthritis: a forty-eight week, multicenter, randomized, doubleblind, placebo-controlled study. *Arthritis Rheum.* 2001; **44**: 1515.
- 25. Quinn CT, Kamen BA. A biochemical perspective of methotrexate neurotoxicity with insight on nonfolate rescue modalities. *J Investig Med.* 1996; **44:** 522.
- 26. Alarcón GS, Tracy IC, Strand GM, *et al.* Survival and drug discontinuation analyses in a large cohort of methotrexate treated rheumatoid arthritis patients. *Ann Rheum Dis.* 1995; **54**:708.
- 27. Schmajuk G, Miao Y, Yazdany J, *et al.* Identification of risk factors for elevated transaminases in methotrexate users through an electronic health record. *Arthritis Care Res* (Hoboken). 2014; **66**: 1159.
- 28. Prey S, Paul C. Effect of folic or folinic acid supplementation on methotrexate-associated safety and efficacy in inflammatory disease: a systematic review. *Br J Dermatol.* 2009; **160**: 622.
- 29. Curtis JR, Beukelman T, Onofrei A, *et al.* Elevated liver enzyme tests among patients with rheumatoid arthritis or psoriatic arthritis treated with methotrexate and/or leflunomide. *Ann Rheum Dis.* 2010; **69**:43.

- 30. Bird P, Griffiths H, Tymms K, Nicholls D, Roberts L, *et al.* The SMILE study -- safety of methotrexate in combination with leflunomide in rheumatoid arthritis. *J Rheumatol.* 2013; **40**: 228-235.
- 31. Laharie D, Seneschal J, Schaeverbeke T, *et al.* Assessment of liver fibrosis with transient elastography and FibroTest in patients treated with methotrexate for chronic inflammatory diseases: a case-control study. *J Hepatol.* 2010; **53**: 1035.
- 32. Saag KG, Teng GG, Patkar NM, *et al.* American College of Rheumatology 2008 recommendations for the use of non-biologic and biologic diseasemodifying antirheumatic drugs in rheumatoid arthritis. *Arthritis Rheum.* 2008; **59**:762.
- 33. Salliot C, van der Heijde D. Long-term safety of methotrexate monotherapy in patients with rheumatoid arthritis: a systematic literature research. *Ann Rheum Dis.* 2009; **68**(7):1100–1104.
- 34. Bird P, Griffiths H, Littlejohn G. Methotrexate in rheumatoid arthritis: efficacy and safety. J Pharmacovigilance. 2014; **2**:127.
- 35. Kremer JM. Toward a better understanding of methotrexate. *Arthritis Rheum*. 2004; **50**: 1370.
- Hider SL, Bruce IN, Thomson W. The pharmacogenetics of methotrexate. *Rheumatology*. (Oxford) 2007; 46:1520.
- Weinblatt ME, Fraser P. Elevated mean corpuscular volume as a predictor of hematologic toxicity due to methotrexate therapy. *Arthritis Rheum*. 1989; 32:1592.
- 38. Gutierrez-Ureña S, Molina JF, García CO, *et al.* Pancytopenia secondary to methotrexate therapy in rheumatoid arthritis. *Arthritis Rheum.* 1996; **39**:272.
- Abásolo L, Júdez E, Descalzo MA, González-Alvaro I, Jover JA, *et al.* Cancer in rheumatoid arthritis: occurrence, mortality, and associated factors in a South European population. *Semin Arthritis Rheum.* 2008; 37: 388-397.
- 40. Mercer LK, Davies R, Galloway JB, Low A, Lunt M, *et al.* Risk of cancer in patients receiving nonbiologic disease-modifying therapy for rheumatoid arthritis compared with the UK general population. *Rheumatology* (Oxford). 2013; **52**: 91-98.
- 41. Menke DM, Griesser H, Moder KG, *et al.* Lymphomas in patients with connective tissue disease. Comparison of p53 protein expression and latent EBV infection in patients immunosuppressed and not immunosuppressed with methotrexate. *Am J Clin Pathol.* 2000; **113**:212.
- 42. Mariette X, Cazals-Hatem D, Warszawki J, *et al.* Lymphomas in rheumatoid arthritis patients treated with methotrexate: a 3-year prospective study in France. *Blood.* 2002; **99**:3909.
- 43. Kamel OW, van de Rijn M, Weiss LM, *et al.* Brief report: reversible lymphomas associated with Epstein-Barr virus occurring during methotrexate therapy for rheumatoid arthritis and dermatomyositis. *N Engl J Med.* 1993; **328**:1317.
- 44. Bernatsky S, Clarke AE, Suissa S. Hematologic malignant neoplasms after drug exposure in rheumatoid arthritis. *Arch Intern Med.* 2008; **168**: 378-381.